Author's response to reviews

Title: Quality of life of Lithuanian women with early stage breast cancer

Authors:

  Giedre Bulotiene (giebul@yahoo.com)
  Jonas Veseliunas (jonas.veseliunas@loc.lt)
  Valerijus Ostapenko (kruties.chirurgija@loc.lt)

Version: 6 Date: 18 June 2007

Author's response to reviews: see over
Dear Sirs,

We are pleased to learn that our manuscript has been accepted by BioMed Central Public Health.

Per your e-mail message received 15 June 2007, we have made the following additions to the paper:

1. We have renamed section ‘Materials & Methods’: ‘Methods’.

2. We have removed tables from the manuscript and placed in the Tables section.

3. We have renamed ‘Figures’: ‘Figure legends’.

Please find the revised manuscript and figure files attached.

Sincerely,

Giedre Bulotiene, MD
Institute of Oncology,
Vilnius University
Lithuania